Author's response to reviews

Title: A Modified Systematic Review of Research Evidence about Education for Pre-registration Nurses in Palliative Care

Authors:

Nahyeni Bassah (nahyeni.bassah@nottingham.ac.uk)
Jane Seymour (jane.seymour@nottingham.ac.uk)
Karen Cox (karen.cox@nottingham.ac.uk)

Version: 2 Date: 2 June 2014

Author's response to reviews:

Nottingham
NG7 2HA
United Kingdom
Nahyeni.bassah@nottigham.ac.uk

Maria Merrie Jul Ladag
Journal Editorial Office
BioMed Central
27/05/2014

Dear Maria,

I will like to submit the attached manuscript “A Modified Systematic Review of Research Evidence about Education for Pre-registration Nurses in Palliative Care” for consideration for publication in the research articles section of BMC Palliative Care.

I have made the following changes to the manuscript as requested by you:
1. I have included page and line numbering to the main manuscript section of the paper.
2. I have included a statement on competing interest after the conclusion.
3. I have also supplied information about authors’ contributions.
4. I have included the figure legend and tables after the references.

Yours sincerely

Nahyeni Bassah
PhD Student, School of Health Sciences,
University of Nottingham, UK